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Reminder from previous week - -
Pump Problems

Two different pumping systems giving us trouble – one at the 
NuMI end (Target Hall) and one at the MINOS end (sump 
area)

Nursing both of them along, trying to make them last until 
the Shutdown.  If either of them breaks, then the Shutdown 
starts for NuMI/MINOS
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Pump Problem #1
Horn-2 water cooling system pumps
Still slowly losing efficiency, but operational for now. 
Jim Hylen implemented his last “trick” on Friday (system using regular + 
backup pump, both working together) and this gained a bit in water 
circulation rate, but not as much as hoped.
When/If this pump system fails, NuMI/MINOS is OFF (no horn cooling)

MINOS Area secondary cooling water loop
Pump Problem #2

Uses cold water from the MINOS sump to cool the primary loops for the 
Beam Absorber and DK pipe, and also the MINOS Hall primary loop, used for 
the magnet coil and electronics racks.
Precise problem not known – FESS and others have been investigating.
When/If this pump system fails, NuMI/MINOS is OFF (no Absorber cooling)
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As long as those 2 pump systems continue to function, we continue 
to follow our Run Plan

Ran Horns-off through Wednesday 2/15, early AM
Horns-on, normal intensity Wed AM – Friday early afternoon
Horns-on, 3-turn medium-low intensity, Friday PM – Sunday AM
Horns-on, normal intensity, Sunday – Tuesday AM
Horns-on, 3-turn medium-low intensity Tuesday – Thursday
Horns-on, 1-turn low intensity to start of Shutdown

Several ES&H accesses for tritium samples in 
addition to machine down-times
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3.4 E18 
Integrated
protons on target
for the week
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Detector up-time during beam

Timing
problem

MINOS
cooling pump
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Protons delivered per week

Integrated Protons on Target for One Week Ending at
Midnight Thursdays
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Total since May/05    1.38 E20 protons


